Author's response to reviews

Title: Operationalizing the RE-AIM framework to evaluate the impact of multi-sector partnerships

Authors:

Shane N Sweet (shane.sweet@mcgill.ca)
Kathleen A Martin Ginis (martink@mcmaster.ca)
Paul A Estabrooks (estabrkp@vt.edu)
Amy E Latimer-Cheung (amy.latimer@queensu.ca)

Version: 5
Date: 2 June 2014

Author's response to reviews:

Dear Implementation Science Editing Team,

RE: MS: 1404012361111360 - Operationalizing the RE-AIM framework to evaluate the impact of multi-sector partnerships

I have accepted all suggested modifications as they did not change the meaning of our manuscript. Please note that I have changed the order of my (Shane Sweet) affiliation as McGill University is my primary affiliation. I have also provided the full institutional address of the authors' affiliations.

Thank you,

Shane N. Sweet, PhD
Assistant Professor
Department of Kinesiology and Physical Education
McGill University
475 Pine Ave W
Montreal, Qc
H2W1S4
shane.sweet@mcgill.ca
514-398-4184 x:09903